Author's response to reviews

Title: Stomatin-like protein 2 is overexpressed in epithelial ovarian cancer and predicts poor patient survival

Authors:

Fei Sun (99609248@qq.com)
Wen Ding (576786473@qq.com)
Jie-Hua He (hejh@sysucc.org.cn)
Xiao-Jing Wang (wangxj@sysucc.org.cn)
Ze-Biao Ma (mazb@sysucc.org.cn)
Yanfang Li (liyf@sysucc.org.cn)

Version: 7
Date: 19 July 2015

Author's response to reviews: see over
Dear Editor,

We appreciate much the comments from Your Handling Editor on our manuscript and have revised the manuscript in accordance with the comments. The following was the point to point responses (please see next page).

Best wishes,

Yanfang Li, Professor,  
Department of Gynecologic Oncology,  
Sun Yat-sen University Cancer Center,  
651 Dongfeng Road East, Guangzhou,  
P.R. China, 510060
Comment 1

1 - Please include a statement in the methods confirming whether written informed consent was required for this study and if not which body waived that requirement.

Response:

We added “A written informed consent from patient was waived by the Committee.” on page 7 140-141.

Comment 2

2 - Please include an Acknowledgements section.

Response:

We added this section on page 20 410-412.